Author's response to reviews

Title: Do positive or negative experiences of social support relate to current and future health? Results from the Doetinchem Cohort Study

Authors:

Simone Croezen (simone.croezen@gmail.com)
H Susan J Picavet (susan.picavet@rivm.nl)
Annemien Haveman-Nies (annemien.haveman@wur.nl)
W M Monique Verschuren (monique.verschuren@rivm.nl)
Lisette C P G M de Groot (lisette.degroot@wur.nl)
Pieter van 't Veer (pieter.vantveer@wur.nl)

Version: 4 Date: 17 January 2012

Author's response to reviews:

We have thoroughly reread the manuscript, and have changed some minor textual errors. Furthermore, we still have some concerns about the tables, as we expressed in our previous communication. We have amended our tables to overcome our concerns.

We have changed the manuscript using the track changes option in MS Word.

Many thanks,
Simone Croezen and Annemien Haveman-Nies